Author's response to reviews

Title: Archetype-based conversion of EHR content models: pilot experience with a regional EHR system

Authors:

Rong Chen (rong.chen@imt.liu.se)
Gunnar O Klein (gunnar.klein@ki.se)
Erik Sundvall (erik.sundvall@imt.liu.se)
Daniel Karlsson (daniel.karlsson@imt.liu.se)
Hans Åhlfeldt (hans.ahlfeldt@imt.liu.se)

Version: 4 Date: 11 June 2009

Author's response to reviews:

Dear editor,

The following changes have been made in this version:
1. All footnotes now are incorporated in the main text.
2. All reference list now contains all authors
3. Reference 1 is changed and provided with all details
4. Place of publicaitons are provided for references 7, 9 and 10. Note that the previous reference 7 wasn't publicly available so it's now replaced by another reference.
5. Links for references 8 and 12 are both tested and seem to be working.
6. Sentence case is used now in the title
7. Vertical elements in the tables are removed